Author's response to reviews

Title: Acquired urethral diverticulum in a paraplegic male presenting with a scrotal mass: A case report

Authors:

Jalal Eddine EL AMMARI (mustaphaahsaini@hotmail.com)  
Mustapha AHSAINI (drahsaini@gmail.com)  
youness Ahallal (mustaphaahsaini@hotmail.com)  
Mohammed Jamal Elfassi (mustaphaahsaini@hotmail.com)  
My Hassan Farih (drahsaini@gmail.com)

Version: 3 Date: 20 September 2012

Author's response to reviews:

Cover letter:

1- All the modification needed was done and highlight by a yellow colors.

2- Written informed consent was obtained from the patients for publication of this manuscript and accompanying images.

3- Language corrections was done by an english teacher from the Americain languages centre (ALC)